Author’s response to reviews

Title: Novel PYGL mutations in Chinese children leading to glycogen storage disease type VI: two case reports

Authors:

Xiaomei Luo (luoxiaomei@xinhuamed.com.cn)
Jiacheng Hu (hujiacheng@sjtu.edu.cn)
Xueren Gao (gaoxueren@xinhuamed.com.cn)
Yanjie Fan (fanyanjie@xinhuamed.com.cn)
Yu Sun (sunyu@xinhuamed.com.cn)
Xuefan Gu (guxuefan@xinhuamed.com.cn)
Wenjuan Qiu (qiuwenjuan@xinhuamed.com.cn)

Version: 5 Date: 20 Mar 2020

Author’s response to reviews:

Dear editor,

We are grateful for the comments of the editor. We have addressed all the comments.

1. Availability of Data and Materials

Due to the time limit, we selected the LOVD database because we were more familiar with it compared to other databases that you recommended. Although we still encountered some problems during data uploading process, the variants information of the two patients (individual ID #00295511 and #00295495) now are viewable to public after communicating with the LOVD curator positively. Please visit the information on the website https://databases.lovd.nl/shared/individuals.

However, part of the information is still under review on the LOVD website. Users could not upload genomic variant data files. We attached the VCF files and emailed to LOVD curator for help although the VCF files are not viewable to the public at present. Anyway, we will keep in touch with the curator of LOVD positively to make the data viewable to the public as soon as possible.
2. Authors’ contributions

In the authors’ contributions section, we have revised the name of the two authors XG.

3. The revised manuscript of clean version has been submitted.

Many thanks for your comments.

Best wishes,

Wenjuan Qiu